Author’s response to reviews

Title: Breath acidification in adolescent runners exposed to atmospheric pollution: a prospective, repeated measures observational study

Authors:

   Jill M Ferdinands (jillferdinands@bellsouth.net)
   Carol A Gotway (cdg7@cdc.gov)
   Roby Greenwald (roby.greenwald@emory.edu)
   David Van Sickle (dvansickle@gmail.com)
   Eric Hunter (eric.hunter@emory.edu)
   W. Gerald Teague (wteague@emory.edu)

Version: 4 Date: 14 February 2008

Author’s response to reviews:

February 14, 2008
Dear Editors,

This revision incorporates editorial suggestions of Dr. David Ozonoff per his email to me on February 4, 2008.

Jill Ferdinands